Author's response to reviews

Title: Effect of interventions to reduce potentially inappropriate use of medicines in nursing homes: a systematic review of randomised controlled trials

Authors:

Louise Forsetlund (louise.forsetlund@kunnskapssenteret.no)
Morten Chr Eike (m.c.eike@medisin.uio.no)
Elisabeth Gjerberg (elisabeth.gjerberg@medisin.uio.no)
Gunn E Vist (gev@kunnskapssenteret.no)

Version: 2 Date: 7 October 2010

Author's response to reviews:

Dear Editor,

How to handle inappropriate medication in nursing homes is a long-standing challenge of international interest. We think that this review of interventions aiming to reduce this problem will fit well into the topics of your journal.

Sincerely,

Louise Forsetlund (corresponding author)
Morten Chr Eike, Elisabeth Gjerberg, Gunn Vist